Author's response to reviews

Title: Clinical categories of patients and encounter rates in primary health care - a three-year study in defined populations

Authors:
Lennart Carlsson (lennart.carlsson@chello.se)
Lars-Erik Strender (lars-erik.strender@neurotec.ki.se)
Gerd Fridh (gerd.fridh@ltblekinge.se)
Gunnar H Nilsson (gunnar.nilsson@nlpo.sll.se)

Version: 3 Date: 1 February 2006

Author’s response to reviews: see over
The final editing of the manuscript (Clinical categories …) have included
- adding nation to the affiliation
- changing headlines in Abstract to Background and Methods including deleting old headings and slight rephrasing of the text accordingly
- Changing headings throughout the manuscript so there are no longer any capitals
- Changing one headline to Methods
- Deleting all indications in the text where to place figures and tables.
- Moving the three pictures away to three separate documents, letting the tables and text and legends of the figures be at the end of the manuscript.